Author’s response to reviews

Title: Impact of a structured multicomponent educational intervention program on metabolic control of patients with type 2 diabetes

Authors:

M Rosario Pinto (mrosariopinto.essaude@gmail.com)
Marta Basto (mlimabasto44@gmail.com)
Lisete Monico (lisete.monico@gmail.com)
Pedro Parreira (parreira@esenfc.pt)

Version: 2 Date: 07 Jun 2017

Author’s response to reviews:

Dear Reviewers

I understand you ask for a publicly accessible registry of this study, and mention that you only accept registration numbers issued by registries that meet all of the ICMJE criteria (http://www.icmje.org/publishing_10register.html). Registries which meet the requirements of the ICMJE include WHO Primary Registries (http://www.who.int/ictrp/network/primary/en/index.html).

When this study started, was only registered on Health Ethics Committee, as demanded by portuguese legislation. It has the registration number 039/CES/INV/2014 and is available in http://www.arslvt.min-saude.pt/pages/504, approved in 06/27/2014.

No other registration was done at that moment, because this registry fulfilled all necessary legal and ethical demands.